Author's response to reviews

Title: A high-throughput high resolution melting protocol for efficient mutation scanning of the BRCA1 and BRCA2 genes

Authors:

Heather L Hondow (heather.hondow@petermac.org)
Stephen B Fox (stephen.fox@petermac.org)
Gillian Mitchell (gillian.mitchell@petermac.org)
Rodney J Scott (Rodney.Scott@newcastle.edu.au)
Victoria Beshay (victoria.beshay@petermac.org)
kConFab investigators (heather.thorne@petermac.org)
Alexander Dobrovic (alex.dobrovic@petermac.org)

Version: 2 Date: 25 January 2011

Author's response to reviews:

This has been transferred intact from Breast Cancer Research who decided not to submit it for review.